Author's response to reviews

Title: Coffee and tea consumption in relation to inflammation and basal glucose metabolism in a multi-ethnic Asian population: a cross-sectional study

Authors:

Salome A Rebello (cmesalo@nus.edu.sg)
Cynthia H Chen (cmechc@nus.edu.sg)
Nasheen Naidoo (cmen@nus.edu.sg)
Wang Xu (a0023748@nus.edu.sg)
Jeanette Lee (ephleej@nus.edu.sg)
Kee Seng Chia (cmedir@nus.edu.sg)
E Shyong Tai (eshyong@pacific.net.sg)
Rob M van Dam (ephrmvd@nus.edu.sg)

Version: 4 Date: 30 May 2011

Author's response to reviews: see over
Dear Nutrition Journal Editors,

We thank you for considering our manuscript “Coffee and tea consumption in relation to inflammation and basal glucose metabolism in a multi-ethnic Asian population: a cross-sectional study” Ref:1889835141513189 suitable for publication in Nutrition Journal.

We have made corrected minor typographical errors. These are
Page 1: Removal of a period sign after Kee Seng Chia
Page 8: Correct from Abbot AXSYM to Abbott AXSYM

Thank you for your help.

Sincerely,

Study authors

Salome A. Rebello, Cynthia H. Chen, Nasheen Naidoo, Wang Xu, Jeannette Lee, Kee Seng Chia, E Shyong Tai and Rob M. van Dam